Author’s response to reviews

Title: Design of the BRISC study: a multicentre controlled clinical trial to optimize the communication of breast cancer risks in genetic counselling

Authors:

Caroline F Ockhuysen-Vermey (karen.ockhuysen@hotmail.com)
Lidewij Henneman (l.henneman@vumc.nl)
Christi J Van Asperen (c.j.van_asperen@lumc.nl)
Jan C Oosterwijk (j.c.oosterwijk@medgen.umcg.nl)
Fred H Menko (fh.menko@vumc.nl)
Danielle RM Timmermans (drm.timmermans@vumc.nl)

Version: 3 Date: 24 September 2008

Author’s response to reviews:

25 September 2008

Re: Submission of revised manuscript by Ockhuysen-Vermey et al.

Dear Editor,

Please find enclosed a revised version of our manuscript by Ockhuysen-Vermey et al entitled: “Design of the BRISC study: a multicentre controlled clinical trial to optimize the communication of breast cancer risks in genetic counselling”, which we would like to submit for publication in BMC Cancer.

Tables have been changed according to the requirements (e.g. using auto format template: simple 1, and excluding the figures from table 1). Figures (n=3) are now presented as separate figure files.

We are looking forward to your decision.

On behalf of the co-authors,

Sincerely,

Lidewij Henneman, PhD, assistant professor, health scientist
(Corresponding author)

Department of Public and Occupational Health,
EMGO Institute
VU University Medical Center
PO Box 7057, 1007 MB Amsterdam
Phone: +31-20-4449815; Fax: +31-20-4448387
E-mail address: l.henneman@vumc.nl